Author's response to reviews

Title: Drug utilization patterns and reported health status in ethnic German migrants (Aussiedler) in Germany: a cross-sectional study

Authors:

Anna Volodina (anna.volodina@yahoo.com)
Thilo Bertsche (thilo.bertsche@uni-leipzig.de)
Karel Kostev (kkostev@de.imshealth.com)
Volker Winkler (Volker.Winkler@urz.uni-heidelberg.de)
Walter Emil Haefeli (Walter.Emil.Haefeli@med.uni-heidelberg.de)
Heiko Becher (heiko.becher@urz.uni-heidelberg.de)

Version: 5 Date: 20 June 2011

Author's response to reviews:

Dear BMC Public Health,

Herewith we would like to submit the final version of the manuscript. The following changes since the last submission have been made:

1. Format of the references has been changed according to journal style;
2. Figure has been cropped as closely as possible to minimise white space around the image;
3. Minor typographical errors have been corrected.